\begin{thebibliography}{10}
\expandafter\ifx\csname url\endcsname\relax
  \def\url#1{\texttt{#1}}\fi
\expandafter\ifx\csname urlprefix\endcsname\relax\def\urlprefix{URL }\fi

\bibitem{Schmidt:01m}
D.~C. Schmidt, R.~Schantz, M.~Masters, J.~Cross, D.~Sharp, L.~DiPalma, {Towards
  Adaptive and Reflective Middleware for Network-Centric Combat Systems},
  CrossTalk.

\bibitem{Gill:01a}
J.~Loyall, J.~Gossett, C.~Gill, R.~Schantz, J.~Zinky, P.~Pal, R.~Shapiro,
  C.~Rodrigues, M.~Atighetchi, D.~Karr, {Comparing and Contrasting Adaptive
  Middleware Support in Wide-Area and Embedded Distributed Object
  Applications}, in: Proceedings of the 21st International Conference on
  Distributed Computing Systems (ICDCS-21), IEEE, 2001, pp. 625--634.

\bibitem{Schantz:04}
P.~Sharma, J.~Loyall, G.~Heineman, R.~Schantz, R.~Shapiro, G.~Duzan,
  {Component-Based Dynamic QoS Adaptations in Distributed Real-Time and
  Embedded Systems}, in: {Proc. of the Intl. Symp. on Dist. Objects and
  Applications (DOA'04)}, Agia Napa, Cyprus, 2004.

\bibitem{ogata97}
K.~Ogata, Modern Control Engineering, Prentice Hall, Englewood Cliffs, NJ,
  1997.

\bibitem{Li:99}
B.~Li, K.~Nahrstedt, {A Control-based Middleware Framework for QoS
  Adaptations}, IEEE Journal on Selected Areas in Communications 17~(9) (1999)
  1632--1650.

\bibitem{Tarek:03}
T.~Abdelzaher, K.~Shin, N.~Bhatti, {User-Level QoS-Adaptive Resource Management
  in Server End-Systems}, {IEEE Transactions on Computers} 52~(5).

\bibitem{Sha:94}
L.~Sha, R.~Rajkumar, S.~S. Sathaye, {Generalized Rate Monotonic Scheduling
  Theory: A Framework for Developing Real-time Systems}, {Proceedings of the
  IEEE} 82~(1).

\bibitem{Schechter:98}
S.~S, M.~Krishnan, M.~D. Smith, {Using path profiles to predict http requests},
  in: Proceedings of the Seventh International World Wide Web Conference (WWW
  98), Brisbane, Australia, 1998.

\bibitem{desidirata:98}
L.~Welch, B.~Shirazi, B.~Ravindran, C.~Bruggeman, {DeSiDeRaTa: QoS Management
  Technology for Dynamic, Scalable, Dependable, Real-Time Systems}, in:
  {Proceedings of the 15th IFAC Workshop on Distributed Computer Control
  Systems (DCCS'98)}, 1998.

\bibitem{Nutt:00}
G.~Nutt, S.~Brandt, A.~Griff, S.~Siewert, T.~Berk, M.~Humphrey, {Dynamically
  Negotiated Resource Management for Data Intensive Application Suites}, in:
  {IEEE Transactions on Knowledge and Data Engineering}, 2000.

\bibitem{Hansen:01}
J.~Hansen, J.~Lehoczky, R.~Rajkumar, {Optimization of Quality of Service in
  Dynamic Systems}, in: {Proceedings of the 9th International Workshop on
  Parallel and Distributed Real-Time Systems}, 2001.

\bibitem{Ghosh:04}
S.~Ghosh, R.~Rajkumar, J.~Hansen, J.~Lehoczky, {Adaptive QoS Optimizations for
  Radar Tracking}, in: {Proceedings of the 10 International Conference on
  Real-time and Embedded Computing Systems (RTCSA'04)}, 2004.

\bibitem{Schmidt:03m}
N.~Wang, D.~C. Schmidt, A.~Gokhale, C.~Rodrigues, B.~Natarajan, J.~P. Loyall,
  R.~E. Schantz, C.~D. Gill, {QoS-enabled Middleware}, in: Q.~Mahmoud (Ed.),
  Middleware for Communications, Wiley and Sons, New York, 2003, pp. 131--162.

\bibitem{Schmidt:05y}
G.~Deng, J.~Balasubramanian, W.~Otte, D.~C. Schmidt, A.~Gokhale, {DAnCE: A
  QoS-enabled Component Deployment and Conguration Engine}, in: Proceedings of
  the 3rd Working Conference on Component Deployment, Grenoble, France, 2005.

\bibitem{Wang:03b}
N.~Wang, C.~Gill, {Improving Real-Time System Configuration via a QoS-aware
  CORBA Component Model}, in: {Hawaii International Conference on System
  Sciences, Software Technology Track, Distributed Object and Component-based
  Software Systems Minitrack, HICSS 2003}, HICSS, Honolulu, HW, 2003.

\bibitem{CCM-light:03}
{Object Management Group}, {Light Weight CORBA Component Model Revised
  Submission}, {OMG Document realtime/03-05-05} Edition (May 2003).

\bibitem{RTCorba:02}
{Object Management Group}, {Real-time CORBA Specification}, {OMG Document
  formal/02-08-02} Edition (Aug. 2002).

\bibitem{Markus:02}
M.~Volter, A.~Schmid, E.~Wolff, {Server Component Patterns: Component
  Infrastructures Illustrated with EJB}, Wiley Series in Software Design
  Patterns, West Sussex, England, 2002.

\bibitem{Schmidt:00a}
D.~C. Schmidt, M.~Stal, H.~Rohnert, F.~Buschmann, {Pattern-Oriented Software
  Architecture: Patterns for Concurrent and Networked Objects, Volume 2}, Wiley
  \& Sons, New York, 2000.

\bibitem{Schmidt:04w}
N.~Wang, C.~Gill, D.~C. Schmidt, V.~Subramonian, {Configuring Real-time Aspects
  in Component Middleware}, in: {Proc. of the International Symposium on
  Distributed Objects and Applications (DOA'04)}, Agia Napa, Cyprus, 2004.

\bibitem{DandC:03}
{Object Management Group}, {Deployment and Configuration Adopted Submission},
  {OMG Document ptc/03-07-08} Edition (Jul. 2003).

\bibitem{Liu:00}
J.~W.~S. Liu, Real-Time Systems, Prentice Hall, New Jersey, 2000.

\bibitem{Das:2005}
B.~Dasarathy, S.~Gadgil, R.~Vaidhyanathan, K.~Parmeswaran, B.~Coan, M.~Conarty,
  V.~Bhanot, {Network QoS Assurance in a Multi-Layer Adaptive Resource
  Management Scheme for Mission-Critical Applications using the CORBA
  Middleware Framework}, in: {Proceedings of the IEEE Real-Time Technology and
  Applications Symposium (RTAS)}, IEEE, San Francisco, CA, 2005.

\bibitem{JunSun:97}
J.~Sun, Fixed-priority end-to-end scheduling in distribured real-time systems,
  Ph.D. thesis, Department of Computer Science, University of Illinois at
  Urbana-Champagne (1997).

\bibitem{Schantz:98}
R.~Vanegas, J.~Zinky, J.~Loyall, D.~Karr, R.~Schantz, D.~Bakken, {Quo's Runtime
  Support for Quality of Service in Distributed Objects}, in: {Proceedings of
  the IFIP International Conference on Distributed Systems Platforms and Open
  Distributed Processing (Middleware'98)}, IFIP, The Lake District, England,
  1998.

\bibitem{Paulo:03}
P.~Stadzisz, J.~Simao, M.~Quinaia, {A Pattern System to Supervisory Control of
  Automated Manufacturing System}, in: {Proceedings of the Third Latin American
  Conference on Pattern Languages of Programming}, Porto de Galinhas, Brazil,
  2003.

\bibitem{Cross:01}
{Joseph K. Cross and Patrick Lardieri}, {Proactive and Reactive Resource
  Reallocation in DoD DRE Systems}, in: Proceedings of the OOPSLA 2001 Workshop
  "Towards Patterns and Pattern Languages for OO Distributed Real-time and
  Embedded Systems", 2001.

\bibitem{Zinky:2002}
J.~Zinky, J.~Loyall, R.~Shapiro, {Runtime Performance Modeling and Measurement
  of Adaptive Distributed Object Applications}, in: {Proceedings of the
  International Symposium on Distributed Objects and Applications (DOA'2002)},
  Irvine, CA, 2002.

\bibitem{Khanna:92}
{Khanna, S., {\em et al.}}, {Realtime Scheduling in SunOS 5.0}, in: Proceedings
  of the USENIX Winter Conference, USENIX Association, 1992, pp. 375--390.

\bibitem{Kenyon:96}
C.~Kenyon, {Best-fit bin-packing with random order}, in: {Proceedings of the
  Seventh Annual ACM-SIAM Symposium on Discrete Algorithms}, 1996.

\bibitem{Baker:85}
B.~S. Baker, {A New Proof for the First-Fit Decreasing Bin-Packing Algorithm},
  in: {Journal of Algorithms}, 1985.

\bibitem{moreira97dynamic}
J.~E. Moreira, V.~K. Naik, Dynamic resource management on distributed systems
  using reconfigurable applications, IBM Journal of Research and Development
  41~(3) (1997) 303--330.
\newline\urlprefix\url{citeseer.ist.psu.edu/moreira97dynamic.html}

\bibitem{Banga:99}
G.~Banga, P.~Druschel, J.~Mogul, {Resource Containers: A New Facility for
  Resource management in Server Systems}, in: Proceedings of the Symposium on
  Operating Systems Design and Implementation (OSDI 99), 1999.

\bibitem{Welch:98}
L.~R. Welch, B.~A. Shirazi, B.~Ravindran, C.~Bruggeman, {DeSiDeRaTa: QoS
  Management Technology for Dynamic, Scalable, Dependable Real-Time Systems},
  in: {IFACs 15th Symposium on Distributed Computer Control Systems (DCCS98)},
  IFAC, 1998.

\bibitem{kyhcm:01:coots}
F.~Kon, T.~Yamane, C.~Hess, R.~Campbell, M.~D. Mickunas, {Dynamic Resource
  Management and Automatic Configuration of Distributed Component Systems}, in:
  Proceedings of the 6th USENIX Conference on Object-Oriented Technologies and
  Systems (COOTS'2001), San Antonio, Texas, 2001, pp. 15--30.
\newline\urlprefix\url{citeseer.ist.psu.edu/341630.html}

\bibitem{shirazi99dynbench}
B.~Shirazi, L.~R. Welch, B.~Ravindran, C.~Cavanaugh, B.~Yanamula, R.~Brucks,
  E.~nam Huh, Dynbench: A dynamic benchmark suite for distributed real-time
  systems, in: {IPPS}/{SPDP} Workshops, 1999, pp. 1335--1349.
\newline\urlprefix\url{citeseer.ist.psu.edu/article/shirazi99dynbench.html}

\bibitem{Schmidt:04z}
K.~Balasubramanian, J.~Balasubramanian, J.~Parsons, A.~Gokhale, D.~C. Schmidt,
  {A Platform-Independent Component Modeling Language for Distributed Real-time
  and Embedded Systems}, in: Proceedings of the 11th Real-time Technology and
  Application Symposium (RTAS '05), IEEE, San Francisco, CA, 2005.

\bibitem{Schmidt:05u}
N.~Shankar, J.~Balasubramanian, D.~C. Schmidt, G.~Biswas, P.~Lardieri,
  E.~Mulholland, T.~Damiano, {A Framework for (Re)Deploying Components in
  Distributed Real-time and Embedded Systems}, in: Poster paper in the
  Dependable and Adaptive Distributed Systems Track of the 21st ACM Symposium
  on Applied Computing, Dijon, France, 2005.

\bibitem{Hill:05}
J.~Hill, J.~Slaby, S.~Baker, D.~Schmidt, {Applying System Execution Modeling
  Tools to Evaluate Enterprise Distributed Real-time and Embedded Systems QoS},
  Tech. Rep. ISIS-05-604, Vanderbilt University (2005).

\end{thebibliography}

% $Id: related.tex,v 1.7 2005/11/23 18:43:01 jai Exp $

\section{Related Work}
\label{sec:relwork}

A significant amount of research on dynamic resource management (DRM)
appears in the literature, spanning several orthogonal dimensions.
For example, DRM capabilities have been applied to specific layers,
such as the application layer~\cite{moreira97dynamic} or network
layer~\cite{Das:2005}.  DRM capabilities have also been applied either
in standalone~\cite{Banga:99} or distributed~\cite{Welch:98}
deployments.  In this section we summarize related DRE research along
the two dimensions.

\iffalse
% Andy, as long as we cite this work earlier let's omit it here!
% OK, I am not sure if we do this in intro or somewhere - andy
The precursor to the multi-layered resource management (MLRM) work in
this paper was described in~\cite{Cross:01}.  That paper described a
pattern language for reactive and proactive resource allocation for
defense applications.  These patterns and principles are at the heart
of the MLRM work described in our work.
\fi

Application-driven distributed DRM was described
in~\cite{moreira97dynamic}, where the authors leverage the
reconfigurability of their parallel applications to dynamically change
the number of executing tasks and allocate resources based on changes
in resource availability.  This work deals with systems in which the
arrival of jobs is unpredictable and highly variable, which is similar
to the TSCE scenarios that are the focus of this paper.  This work,
however, is application-specific and provides a point solution that is
not based on standard middleware platforms.  In contrast, our ARMS
MLRM work provides standards-based reusable middleware that are
designed to support a broad class of enterprise DRE systems that can
benefit from DRM.

The work described by~\cite{kyhcm:01:coots} presents an integrated
architecture for automatic configuration of component-based
applications and DRM for distributed heterogeneous environments.  The
authors, however, do not describe how end-to-end QoS capabilities are
maintained for mission-critical applications.  In contrast, our work
provides the results of empirical benchmarks that evaluate how well
our MLRM supports end-to-end QoS.

Early work on middleware-based DRM solutions in the context of DRE
shipboard computing systems is described in~\cite{Welch:98}, where the
authors describe the DeSiDeRaTa middleware that provides DRM for
QoS-sensitive applications.  This work and related efforts
in~\cite{shirazi99dynbench} describe the characteristics necessary for
DRM and provide the conceptual foundations for much of the ARMS MLRM
efforts.  Our work on MLRM extends this earlier work to support more
finer grained DRE capabilities that ties mission needs to the
management of physical resources, while also allowing for capability
upgrades.

Network layer DRM capabilities conducted in the DARPA ARMS program are
described in~\cite{Das:2005}.  The network layer DRM solution in that
paper focus on realizing DRM capabilities for one layer (the network
layer) and for a specific resource (bandwidth).  The MLRM approach
described in this paper extends the network layer DRM capabilities to
manage multiple resources, including CPU and network bandwidth,
thereby providing a more comprehensive multi-layered and end-to-end
solution.

%%% Local Variables: 
%%% mode: latex
%%% TeX-master: "JSS-2006"
%%% End: 

